

E-Textiles for Sports and Fitness Sensing: Current State, Challenges, and Future Opportunities

Kai Yang ¹, Stuart A. McErlain-Naylor ², Beckie Isaia ³, Andrew Callaway ⁴ and Steve Beeby ^{3,*}

- Winchester School of Art, University of Southampton, Southampton SO23 8DL, UK; ky2e09@soton.ac.uk
 School of Sport Everying and Health Sciences, Loughborough University, Loughborough UE11 2TU, UK;
- ² School of Sport, Exercise and Health Sciences, Loughborough University, Loughborough LE11 3TU, UK; s.a.mcerlain-naylor@lboro.ac.uk
- ³ Centre for Flexible Electronics and E-Textiles (C-FLEET), School of Electronics and Computer Science, University of Southampton, Southampton SO17 1BJ, UK; r.j.isaia@soton.ac.uk
- ⁴ Department of Rehabilitation and Sport Sciences, Bournemouth University, Bournemouth BH12 5BB, UK; acallaway@bournemouth.ac.uk
- * Correspondence: spb@ecs.soton.ac.uk; Tel.: +44-2380-596-663

Abstract: E-textiles have emerged as a fast-growing area in wearable technology for sports and fitness due to the soft and comfortable nature of textile materials and the capability for smart functionality to be integrated into familiar sports clothing. This review paper presents the roles of wearable technologies in sport and fitness in monitoring movement and biosignals used to assess performance, reduce injury risk, and motivate training/exercise. The drivers of research in e-textiles are discussed after reviewing existing non-textile and textile-based commercial wearable products. Different sensing components/materials (e.g., inertial measurement units, electrodes for biosignals, piezoresistive sensors), manufacturing processes, and their applications in sports and fitness published in the literature were reviewed and discussed. Finally, the paper presents the current challenges of e-textiles to achieve practical applications at scale and future perspectives in e-textiles research and development.

Keywords: e-textiles; wearable technology; sensors; sports; fitness; performance; injury; monitoring; rehabilitation

1. Introduction

Wearable technologies are now accepted and widely used in multiple sports and fitness activities across all levels of performance, from recreational to elite, in individual and team sports, and including non-disabled and disabled athletes alike [1]. Wearables can be used to monitor a wide variety of biosignals (e.g., heart rate and muscle excitation) and can also track performance (e.g., distance and speed) and the technique (e.g., joint angles) used to produce that performance. The most common types of existing wearables are typically wrist-worn smartwatches, chest straps, devices mounted on or in the footwear, or, more recently, those located within sports clothing [2]. Analysis of subsequent data can be used to gauge improvements in fitness, help mitigate injury risk [3,4], inform recovery [5], monitor technique [6], or, at a consumer level, simply provide motivation [7]. The typical sensors used to date are inertial measurement units (IMUs comprising accelerometers, gyroscopes, and magnetometers), Global Positioning Systems (GPS), and heart rate (electrocardiography, ECG) and muscle excitation (electromyography, EMG) sensors.

Electronics textiles (e-textiles or smart fabrics) are advanced textiles that include electronic functionality ranging from conductive tracks to sensing/actuating, communications, and microprocessing [8]. The global market for e-textiles is projected to reach around \$15 billion by 2028 [9]. E-textiles are a platform technology for wearables that are highly relevant to sports and fitness applications. The motivation for incorporating sensing and electronic functionality into textiles for sports and fitness applications is evidenced by the growing number of clothing-based wearable devices aimed at this sector. Examples include



Citation: Yang, K.; McErlain-Naylor, S.A.; Isaia, B.; Callaway, A.; Beeby, S. E-Textiles for Sports and Fitness Sensing: Current State, Challenges, and Future Opportunities. *Sensors* 2024, 24, 1058. https://doi.org/ 10.3390/s24041058

Academic Editor: Eduardo García Breijo

Received: 23 December 2023 Revised: 23 January 2024 Accepted: 30 January 2024 Published: 6 February 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). STATSports Apex Athlete series [10] and Catapult One [11], both of which are GPS-based tracking devices also incorporating IMUs, where the electronics containing sensing, communication, and processing capabilities are implemented in a conventional rigid form with the modules located in a pocket on the garment. GPS-enabled smart watches or footwear could provide similar GPS data; however, their use is not permitted in most sports where physical contact between participants is possible, but this is dependent on the changing rules of each sport.

Textiles provide a comfortable, ubiquitous platform that individuals are entirely familiar with. Considerable effort has gone into engineering technical textiles for sportswear [12], where the market is dominated by household brands such as Nike, Adidas, and Puma. E-textiles technology offers the ability to further enhance sportswear functionality by invisibly integrating sensors, microprocessors, and communications into garments [13,14]. This approach can potentially improve compliance with technology amongst users, develop the ecological validity of the data where sensing can happen in the natural sporting environment, and collect data on more important metrics about populations remotely to develop sensing algorithms and interventions. However, at present, the level of integration of the electronic functionality within the garment is typically limited to separate modules that fit into a pocket located on the clothing, as illustrated by the STATSports and Catapult cases. A few other examples do include additional functionality in the textile. Prevayl's Smartwear incorporates conductive textile electrodes connected to the electronic unit with a 512 Hz sampling frequency to detect the ECG signal and display the information on a smartphone (Figure 1a) [15]. The product is used for both amateur and professional athletes. Equivital LifeMonitor with built-in multi-sensors can monitor ECG, respiration, tri-axis accelerometry, and temperature (Figure 1b). The product is mainly used in training (e.g., military) and health and safety monitoring in harsh working environments (e.g., fire-fighting), but it can also be used in professional sports monitoring (e.g., car races, bike races) [16]. Despite the advances in e-textiles, it is not straightforward to achieve the required reliable and robust electronic and sensing capability in textiles in a manner that has minimal effect on the properties of the fabric. Most existing e-textile technology does not yet deliver practical solutions that replicate the levels of sensing, processing, and communication functionality achieved with the separate, rigid, discrete modules located in a pocket within the garment.



Figure 1. (a) Prevayl ECG vest [15], reproduced with permission from Prevayl; (b) Equivital multiparameter monitoring system [16], reproduced with permission from Equivital.

This review contains a comprehensive evaluation of the capability of existing technologies and approaches to addressing a wide range of sports/fitness-related applications and looks forward to the development of e-textiles and the corresponding research challenges. Section 2 introduces the role of technology in sports in general, reviews the measurements and parameters that are of interest to the participant or coaching/support team, and highlights a use-case scenario of monitoring training loads in an attempt to promote beneficial adaptations and reduce injury risk. Section 3 discusses the commercial non-textile and textile-based wearable devices in sports and fitness applications. Section 4 outlines the drivers for research in e-textiles and presents research-based examples of existing e-textile systems using different sensing technologies. Section 5 discusses future opportunities and the associated research and development required to realise practical e-textile solutions that assist individuals across all abilities and age ranges.

2. Technology and Sensing in Sports and Fitness

2.1. Current State and Future Considerations of Technology in Sports Science

Technology has always played an important role in sports science; without it, much of the scientific foundations within its sub-disciples would not have been possible. Historically, sports science was often restricted to a laboratory-based setting, in subject discipline isolation, but the rapid development of technology (sensors, processing, and communications) has allowed the movement from the laboratory to training and competition venues and environments. Calls to enhance the interdisciplinary nature of sports science work [17–21] have slowly been realised in research [22–26], but the call is ever-present. This multidisciplinary approach has happened more rapidly in practice in tandem with the professionalism of multidisciplinary teams [27] involving coaches, strength and conditioning specialists, medical doctors, rehabilitation therapists, physiologists, psychologists, nutritionists, and many others. This has likely happened with the commercial development of technology (e.g., cameras, heart rate monitoring, mobile force platforms), allowing the capture of a variety of data to measure athletic performance and facilitate the design of interventions to enhance performance, recover from injury, and monitor wellness, sleep, and diet.

Athletes wear smartwatches, fitness trackers, heart rate monitors, and other sensors to track their performance, monitor their health, and analyse their training. Previously outlawed in many competitions, there is a slow but welcome allowance to wear sensing technology in competitions, although sport-dependent. These devices can provide realtime feedback metrics such as heart rate, calories burned, and distance covered, helping athletes optimise their workouts and avoid overtraining. They can also help coaches track the progress of athletes they work with and make informed decisions about training and competition.

As wearable devices become more advanced, for example, with the development of communication to transfer more information more rapidly and with the development of mathematical processing from inertial sensors, they will provide ever more detailed and accurate data, giving athletes and coaches a deeper understanding of physical and tactical performance, and training loads. Analytics of this data will continue to improve, allowing analysts to identify more complex patterns and trends and make more accurate predictions about athlete performance and injury risk. However, many these processing capacities will lie with large companies interrogating ever-growing data sets.

Virtual and augmented reality are a growing trend in sports [28–30] and have the potential to have a large impact on how we learn skills [31] as they become even more realistic, providing athletes with an immersive training experience that closely mimics the real thing. However, as technology becomes more integral to sports performance, there is a risk that it could create a divide between well-funded teams and athletes and those who cannot afford the latest equipment and technology. This could widen the gap between elite athletes and those who are just starting out, making it even harder for newcomers to break into the sport [32–34]. At the same time, there is an opportunity to develop interventions using virtual reality to develop the talent pool of athletes available, as long as practitioners are sufficiently trained in how to use and adapt said interventions.

From an athlete/coach/practitioner perspective, we already have a deluge of data, often with more metrics and data recorded than can be actioned. Some simple but key questions need to be asked as we progress with these technological developments. Are the metrics measured valid and reliable across a range of populations? As we move from laboratory to different training environments, how rigorous does the setup need to be? As technology develops different metrics, how sensitive are these to change (e.g., minimum clinically or practically significant change or minimum detectable change), and what does this mean for different populations? Are metrics named for trademark and marketing

terms obscuring good science or confusing users? Do the users understand what this metric is and how it can be modified through intervention—is it useful? Future academic work will need to focus on developing actionable interventions through knowledge transfer with companies and teams to develop an understanding of training interventions. With a global pandemic of physical inactivity leading to early mortality and mental health conditions [35], there is an opportunity, and a need, to develop technology to enhance our ability to personalize health, rehabilitation, and well-being, which will start to be intervention-led in conjunction with the progression of technology.

2.2. Technology for Training Load Monitoring in Sport

A common application of technology within sports is for training load monitoring. Whether physical activity is performed for health and/or social benefits, sporting performance, injury risk reduction, and/or post-injury rehabilitation, a primary concern is the prescription and monitoring of an appropriate 'dose' of activity. This is often described as the 'training load', representing the demands of the particular activity for the particular individual in the particular context that it was performed [36]. At a simplistic level of understanding, there is hypothesised to exist a 'sweet region' of ideal training load, below which prior adaptations may be lost (i.e., disuse) and above which the negative consequences (e.g., tissue damage) of activity may exceed any beneficial adaptations (i.e., overuse) [37,38]. Additionally, the balance between training stimulus and inter-session recovery feeds into a 'fitness-fatigue' model of performance [39]. These simplistic models can be applied to various biological systems and their related performance and/or injury effects [40].

Training load is often divided into 'external' and 'internal' load. External load refers to the activity and work completed (e.g., distance travelled, mass lifted, number of repetitions), whereas internal load refers to the effects of that activity on biological systems [41–43]. In this context, 'load' may not necessarily refer to the mechanical force experienced but rather a description of the demands of the activity, just as in familiar terms such as 'workload', 'cognitive load', or 'viral load' [44,45]. Each of these categories can be further subdivided into physiological or biomechanical loads. For example, external physiological load may refer to metabolic power, whereas internal physiological load could be the oxygen uptake or cardiovascular demand [46]. These metrics are typically measured using laboratory equipment where a participant would participate in a VO2max (or equivalent) test to measure oxygen uptake or take lactate samples from blood to calculate fatigue levels. Through experimentation, regression equations [47,48] have been created to allow estimates of VO2max, which can then be calculated using input from worn sensors (e.g., heart rate), yet lactate has only recently become non-invasive [49] showing potential for e-textiles in the future.

External biomechanical load may refer to ground reaction forces or centre of mass accelerations, whereas internal biomechanical load may refer to joint contact forces or muscle-tendon forces [44]. These alternate loads can, therefore, be prescribed, monitored, and altered somewhat independently of each other to achieve the overall aims of a training block or individual session, perhaps while also addressing secondary aims relating to injury risk management [50].

Perhaps the most well-known and commonly used technologies for monitoring training load in sports, as already alluded to within this article, are heart rate monitors and GPS or Global Navigation Satellite Systems (GNSS) technologies. GPS is one sub-section of GNSS, with modern GNSS accessing a greater number of satellites and, therefore, potentially greater precision and reliability [51,52]. Both are typically monitored at the torso, heart rate via a chest strap, and GNSS via a unit positioned between the scapulae in a manufacturer-provided elastic harness. Heart rate data are used to calculate metrics such as peak or mean heart rate or time spent within heart rate zones generally corresponding to the use of different physiological energy pathways (e.g., aerobic/anaerobic energy systems). GNSS, on the other hand, can be used to calculate metrics, such as total distance covered, number of sprints, top speed, work:rest ratios, and the time spent in different speed zones. Such training load metrics can not only be used for the prescription and monitoring of training sessions but can also facilitate manual or automated updating of subsequent session designs based on previous differences between prescribed and achieved training loads [53].

One area of growing focus is the need and current relative inability to measure biomechanical training loads outside of a laboratory [45], especially internal biomechanical training loads. That is, the forces experienced by specific tissues within the body (e.g., muscle, bone, tendon, ligament). This is particularly relevant when seeking to associate training load measures with injury likelihood or inform progressive overloading of the tissue during rehabilitation or injury risk reduction programs. GNSS can describe the activity performed (e.g., a 10 km run at a certain speed), and heart rate can indicate the cardiovascular demands of the activity, but neither can describe the effects of the exercise on the musculoskeletal system. Applications of GNSS are further limited in indoor sports (although Local Positioning Systems can be used [54,55]) and when activity is performed with relatively little translational movement (e.g., holding an isometric squat position in a badminton doubles match). As a result, many companies now incorporate an IMU into the GPS unit, with the accelerometers used to measure cumulative accelerations experienced at the back. These are expressed as modified load vectors such as PlayerLoadTM (Catapult Sports, Melbourne, Australia [56,57]) or Dynamic Stress Load (STATSports, Newry, Northern Ireland [58,59]). As the trunk is the heaviest body segment and is positioned proximally on the body, these accelerations have been proposed to represent whole-body mass centre accelerations and, therefore, relate to the ground reaction forces or impact magnitudes experienced by the athlete (i.e., external biomechanical training load). However, this neglects the influence of other body segment accelerations, considerable post-impact shockwave attenuation inferior to the sensor [60], and the contribution of various frequency components to the overall acceleration signal [58,61]. Even an entirely accurate measure of ground reaction force or mass centre acceleration may not correlate with the internal forces experienced by specific tissues if muscle forces are not accounted for [62]. Nonetheless, overall measures of cumulative torso accelerations have been associated with oxygen uptake and heart rate measures [54,63], distinguished between activities [64], and enabled the monitoring of fatigue [56], among other factors.

IMUs positioned elsewhere on the body offer additional measurement opportunities. IMUs on specific body segments can be used not only to quantify technique but also in conjunction with machine learning or other algorithms used to detect certain activities [65–68] and perhaps quantify the intensity of these movements. Greater insight can perhaps be gained when data from wearable technologies are used alongside existing models of tissue stress response or as inputs to neuromusculoskeletal models. For example, models of bone remodelling [69,70] have been used to quantify a tibial 'Bone Stimulus' metric from tibial accelerometry via IMeasureU's IMU Step system. While this has shown reliable results in running and soccer-related tasks [71,72], studies have demonstrated an inability of tibial accelerometry to represent loads on the tibia bone [61,73]. Such bone forces are more dependent upon muscle forces than any shockwave resulting from the ground reaction force [61]. Nonetheless, physics-based and/or machine-learning models can be used to predict tibial bone loads from wearable technology data (e.g., a pressure-sensing insole and an IMU positioned on the foot) [74,75]. Similarly, open-source methodologies exist for the use of multiple IMUs as input to whole-body musculoskeletal computer simulation models that could also be used to estimate internal biomechanical loads [76,77]. However, the number and/or size of sensors required will likely need to be reduced before any real-world application for training load monitoring, perhaps predicting unknown segment kinematics via machine learning [78–80] or optimal control algorithms (i.e., determining the objective or 'cost function' of an individual's movement via inverse optimal control and using this to predict unknown kinematics during the movement) [81–83]. These predictions and the selection of optimal sensor positioning and processing may be augmented through the use of synthetic wearable technology data sets based on real or augmented laboratory

motion capture data sets [73,84–88]. Alternatively, it may be possible to identify algorithms to recognize certain activities [64–67] and then scale tissue-specific load estimates from laboratory-based estimates using an intensity measure derived from a reduced number of sensors [89]. The use of e-textile solutions with greater comfort and reduced conscious athlete awareness may somewhat remove the constraint on the number and position of sensors. Any developed tools for the estimation of tissue-specific internal biomechanical training loads may facilitate the prescription and monitoring of specific loads, as well as the ranking of exercises in terms of specific tissue loading [90,91] or the provision of haptic [92,93] or audio [94,95] biofeedback.

There are a number of important considerations when choosing or developing wearable technology for sports, particularly for biomechanical loads. These include hardware mass, dimensions, fixation, sensor range and sampling frequency, and calibration routines [96]. If the technology is intended to be used for injury-related applications, then it should additionally build upon established causal relationships, be applicable without any laboratory-based inputs, and be informed by specific guidelines such as individualor population-based normative boundaries, thresholds, or trends [97]. Researchers and manufacturers should not forget the context in which the technology is to be applied and the needs and preferences of the user (e.g., athlete and coach) [98]. Finally, we should assess not only the metric itself but also the consequences of prescribing changes in that metric, especially when the metric being used relies on average relationships rather than a direct measure of physiological and/or biomechanical loads.

As stated in a recent International Society of Biomechanics in Sports career award paper, 'In the future... training and rehabilitation programmes will use wearable and simple imaging technologies to estimate tissue level biomechanics derived from personalised neuromusculoskeletal modelling in real-time in the real-world. The future is not that far away' [99]. Developments in e-textile technologies may help to make that future a reality.

3. Commercialised Wearable Technologies for Sports

A wide variety of wearable technologies have been developed for or used in elite sports and consumer fitness-related applications. The following are examples of commercial devices, including where they have been explored in the scientific literature.

3.1. Non-Textile/Clothing-Based Wearables

Wrist-based wearables are the most popular form of wearable technology for amateur athletes and have been used in numerous studies [100]. There are too many commercial smartwatches available to list here, but examples such as the Apple Watch and Fitbit Charge series provide typical functionality associated with such devices. This includes heart rate monitoring, general activity tracking (steps, distance, energy expenditure, floors climbed, built-in GPS) and recognition and tracking of particular sports/activities (e.g., walking, running, cycling, swimming). A comprehensive list of devices and their applications, together with a summary of performance evaluation, has been presented by Cosoli et al. [101,102]. Most smartwatches incorporate optical heart rate measurement by photoplethysmography (PPG). This approach can be subject to motion artefacts, and the quality of the fit and location on the wrist can also affect measurements, especially when active [103]. Error rates are higher when swimming, where arm movements and the water can also affect the PPG-based sensors [104]. Where arm movement is a key aspect of a particular sport (e.g., baseball pitching or tennis serving), the IMUs within smart watches can obtain sufficient kinematic data to provide real-time feedback on arm-related technique, allowing the player to improve their performance, leading to an improvement in ball speed and the pronation movement in serve [105]. The wearable WHOOP wristband is a wristband specifically focused on fitness and health that tracks an individual's training, recovery, and sleep [106]. It uses PPG sensors to monitor heart rate, heart rate variability, and sleep, and it uses this data to estimate training magnitude whilst exercising and the rate of recovery. Other specialised wrist-based sensors have been developed for particular

sports. For example, the BeastTM wrist-based sensor comprises IMUs with the associated data processing and software, providing information tailored to resistance training [107]. The system provides information on repetition speed and power but was found to be less accurate than linear position sensors when measuring speed, which could be due to the position when worn [108]. Ultimately, the location on the wrist is not ideal for monitoring many parameters and measurements are inferred, leading to higher error rates compared to sensors positioned optimally in terms of performance. For example, straightforward measurements such as step counting can be inaccurate compared with foot-based sensors.

Another commonly used and well-established class of wearable devices for monitoring users when undertaking physical activity is chest straps with heart rate detection. These detect the electrical signal associated with a heartbeat through the skin, and hence, the electrodes located on the strap must be in contact with the skin. The same technique is used for clinical ECG measurements, although chest straps are limited to 2 electrodes, whereas clinical systems can use over 20. Early research into their efficacy was very positive, demonstrating that consumer sports chest straps achieve 99% accuracy when compared to clinical ECG equipment [109]. Studies exploring the suitability of chest straps to provide data enabling measurement of the RR interval (the time between successive R waves in an ECG signal) found that at high activity levels, the Polar H10 chest strap was superior to a Holter device [110]. The Holter device is commonly used as a reference but was shown to be unsuitable for intense activities and large body movements, whereas the chest strap remained unaffected.

Wearable inertial sensors can be located on the body using pockets in clothing or by simply strapping the sensor onto, for example, a limb. These started with pedometers recording step count/frequency for daily ambulatory monitoring. The limitation of these is that only total steps for the duration of the recording are shown, and no temporal information is provided for the calculation of the rate of change of steps (i.e., were they running or walking). As technology has progressed, inertial sensor-based devices have helped address this limitation, allowing many sports to obtain various spatiotemporal parameters [111].

The TritonWear (first version) system for monitoring swimmers comprises the Triton Unit (firmware version 1.1.2), a small IMU designed to fit under a swimmer's cap, and data analysis and presentation software. The system can monitor up to 50 swimmers, providing general motion data such as acceleration and swimming-specific data (e.g., stroke count and stroke efficiency) with reported mean absolute percentage error (MAPE) of 0, 2.4, 7.1 and 4.9% for butterfly, breaststroke, backstroke & freestyle respectively for stroke count data [112].

The Xsens MVN Awinda is a real-time human motion tracker that exploits IMUs placed at multiple locations around the body using simple Velcro straps. The devices are connected with a bespoke wireless protocol that enables synchronised data capture [113]. This system has been used to analyse technique during walking [114,115], running [116,117], cycling [118,119], and more novel activities such as goalkeeper diving [120], cricket bowling [121], change of direction tasks [122], and jumping in handball [123]. A similar array of sensors, with the addition of EMG sensors designed to be stuck to the skin, has been developed by dorsaVi Ltd. (Victoria, Australia) [124]. These sensors can be mounted at the optimum position for the activity type and have been developed for elite sports, medical and workplace applications (e.g., to monitor lifting). The inertial sensors have been validated when placed on the medial tibia of each leg and used to predict the ground reaction forces when running. The system achieved errors in the range of 5.4–6.1% across three participants when compared to a force platform [125]. The dorsaVi technology produces comparable kinematic data to the Xsens Awinda system, indicating that the different mounting methods have little effect on the data collected [126].

3.2. Clothing and Textile-Based Wearables

IMUs located within clothing have been developed for a range of specific sports. For example, sensors placed along the spine can be used to identify swimming stroke and phase [127], commercialised with the Incus Nova wearable [128]. This sits in a pocket between the shoulder blades and can also provide data when running. This highlights the broad applicability of IMUs, where the same hardware can provide data for each activity type, subsequently requiring the appropriate analysis. The STATSports Apex Athlete series [10] and Catapult One [11] GPS devices are designed to monitor an athlete's performance in field sports where parameters such as the distance covered, number of sprints, time in different speed zones, and top speed are monitored. These garment-based wearables allow the electronics to be placed on the upper back between the scapulae, where they are safe and relatively unobtrusive. The STATSports Apex, for example, has been compared favourably with radar-based tracking technologies [129] and has been used to gain insight into the performance of football players in different age groups, the results of which could inform training programs [130]. The Nadi X, developed by startup company Wearable X [131], include accelerometers and haptic feedback in the form of vibrations that are designed to assist with obtaining and maintaining yoga positions. The rigid device mechanically mounts to compression-fitting yoga pants behind the knee.

In the previous examples, the textile itself has no functionality other than to include a pocket or mounting point for the conventional rigid electronic modules. In addition to Prevayl's Smartwear mentioned in the introduction with ECG electrodes embedded in garments, other manufacturers have also increased the functionality of the textile or garment. ZephyrTM has developed tight or loose-fitting t-shirts and a sports bra that incorporate heart rate and heart rate variability sensing electrodes and a mounting point for a detachable rigid module that monitors breathing and activity via IMUs. The garments are essentially a chest band incorporated within the shirts/bra. The sensors utilise silver conductive fabric electrodes that must be moistened before use, and the module must be removed for washing and recharging [132]. Another smart garment developed by Hexoskin also includes textile electrodes within the close-fitting ProShirt that provide cardiac, respiratory, and activity monitoring [133]. The associated electronics, IMUs and power supply are again provided in a rigid electronic module that sits within a pocket in the garment. The validity and reliability of the Hexoskin smart shirt for measuring heart rate during strenuous physical activity have been evaluated by comparing results with a Polar Team Pro chest strap [134]. During peak activities with multidirectional upper body movements, the Hexoskin provided erroneous data due to motion artefacts in 4 of the nine athletes studied. Another vest-mounted heart rate monitoring top with two textile electrodes is the Equivital Lifemonitor, but this, too, is subject to motion artefacts and is better suited to lower activity levels [135]. These examples highlight motion artefacts as one of the major challenges with wearable ECG/EMG sensors when used in physical activities. Sensoria offers short-sleeved sports tops and bras that use textile electrodes, a snap-on electronics module, and a Smart sock that utilises three textile pressure sensors under the foot to capture cadence, step count and foot landing technique. The sensors are connected to the Sensoria Core, a small, rigid electronics module that snaps into the dock located at the top of the sock. In comparison with a shoe-based inertial sensor and video analysis, the smart sock provided an excellent measure of cadence (ICC 0.91), speed (ICC 0.86), distance (ICC 0.86) and foot strike pattern (ICC 0.91) [136]. Fabric-embedded electrodes are used in the Athos Core compression garment designed to monitor muscle excitation alongside heart and breath rates. The data collected by the Athos system were found to be comparable with a research-grade surface EMG system [137]. A similar product developed by Strive includes an IMU attached to compression-fitting shorts that include electrodes for detecting EMG signals from quadriceps, hamstrings, and gluteal muscles [138]. A study funded by Strive reported a good correlation between muscle and training loads [139]. Dragonfly Golf is a clothing-based all-body motion analysis system that comprises 18 IMUs attached to a full-body base layer garment [140]. The IMUs are housed in rigid modules that fasten

to the corresponding connector on the suit, and this enables sensors to be removed for washing and then re-attached. The suit includes a wiring loom that connects the sensors to the central power supply, and a module that collects and wirelessly transmits the data. The hardware is complemented by player and coaching apps that enable the visualisation of the data and detailed analysis of the golfer's technique.

E-textiles do not have to be engineered into full garments. For example, smart sensing sleeves that cover the forearm or full arm have been developed. The Komodo AIO Smart Sleeve is a general-purpose wearable device that incorporates a PPG sensor, IMU, battery and Bluetooth Low Energy hardware in a typical rigid module that snaps onto the inside of the sleeve and sits in contact with the skin [141]. The sleeve is available in both short (lower arm only) or long (full arm length) versions, and in active mode, the device monitors steps, distance, heart rate, sleep, and activity intensity. In health mode, the sleeve has an additional wired electrode that connects to the module and is designed to be placed on the chest. This provides one lead, two electrodes, and ECG data when stationary and cannot be used when training. The level of functionality within the textile is low, with the sleeve simply incorporating magnetic clip-on connectors and a single embedded wire to the top connector. Motus has developed a sleeve for monitoring throwing sports such as baseball, American football and cricket, but the functionality is again provided by a rigid module that sits within pockets on a sleeve or band [142]. Myontec offers a suite of garments, including compression shirts, shorts, waist belts and arm and lower-leg sleeves with IMU sensors and EMG electrodes [143]. The shorts provide similar functionality to the Stive system, with EMG sensing electrodes monitoring the quadriceps, hamstrings, and gluteals. This has been used to longitudinally analyse neuromuscular responses to training [144]. The leg sleeve monitors the tibialis, gastrocnemius and soleus muscles, whilst the belt targets the multifidus and erector spinae muscles in the back. The same module housing the IMU, and other EMG monitoring electronics has been used within each garment using a bespoke snap-on rigid connector.

This review highlights that many of the commercially available wearables, clothingrelated or otherwise, rely on IMUs potentially augmented with GPS or heart rate monitoring systems. The hardware and sensor performance offered is similar across all examples, with some variations in the number of sensors and the method/location of attachment. An additional and key distinction is provided in the supporting application software that collects and analyses the data. It is a crowded marketplace, and despite the market size for wearable technologies and the expected growth, successfully monetising the technology and surviving the competition is challenging [145].

4. Research in E-Textiles for Sports and Fitness

4.1. The Motivation for E-Textiles Research on Sports and Fitness

Textile implementations of sensors and electronics have a large range of potential applications in monitoring progress and performance, reducing injury risk and motivating regular exercise and training. To advance the current commercial products and enable the wide adoption of e-textile technologies in sports and fitness, research has been conducted in the areas of new materials and sensor components, manufacturing methods, system integration and user-centred design. The aims of these studies include improving sensing functions (e.g., accuracy, reliability), user comfort (e.g., flexibility, softness), durability (e.g., washing, wearing) and design (e.g., aesthetic appearance, ease to put on and take off). Other research has related to the charging/power, data acquisition/processing/transmission and the interaction with other technologies (e.g., Internet of Things (IoT), Artificial Intelligence (AI)), which are not within the scope of this review but for which information can be found in published papers [8,146,147].

The benefits offered by e-textile implementation depend upon the application and the solution, but there are several common advantages:

- Textile/clothing-based solutions provide a comfortable and familiar platform to users.
 E-textiles would enable unobtrusive and ubiquitous deployment sensors in clothing.
 Textile-based soft products are safe in contact sports;
- Textiles are versatile materials that can be designed to change their properties to fit the application needs through the optimal combination of textile materials and structures;
- The integration in clothing will improve compliance (users might forget to use conventional technology, but they always remember to get dressed). Ease of use and increased compliance can provide more data to better inform training and reduce the risk of overload;
- The unobtrusive and seamless integration of miniaturised or flexible sensors is less likely to influence the parameters being monitored. This allows close contact between the sensors and the skin, reducing measurement errors caused by the displacement of the rigid/large sensors relative to the underlying anatomy;
- Multiple sensors can be incorporated into a single platform (e.g., an item of clothing) in multiple body positions rather than requiring users to wear several separate devices;
- Integration of sensing within garments enables sensors to be located at the optimal location on the body and to measure a much wider range of signals than possible with, for example, a smartwatch;
- Information or alerts can be provided through the textile, providing real-time feedback to the user in a single platform.

4.2. Research on E-Textiles for Sports and Fitness Using Different Sensing Technologies

IMUs, biopotential electrodes and tactile/pressure interfaces are the most commonly used technologies in sports [148]. Main applications include motion measurement, vital signs monitoring and interactive applications. The level of sensor-textile integration varies from simply attaching the sensor to the wearer using a textile (e.g., elastic strap) to integrating the sensor components (e.g., IMU chip, sensing yarn/ink) to form an integral part of the wearable e-textile.

4.3. IMUs for Motion Detection and Joint Function Measurement

The use of IMUs has accelerated with improvements in hardware development and data processing. IMUs have been used in e-textile wearables for sports and fitness due to their low cost, portability, real-time data, and suitability for dynamic movements. They have been used to monitor gestures/poses, range of motion, steps and types of activities (e.g., walking, running, climbing). They can also be used in gait analysis to assess movement patterns and monitor rehabilitation. IMUs have shown moderate to excellent correlations with gold standard approaches for gait spatiotemporal parameters during running [149], but users should be aware of poorer accuracy in high-speed and non-sagittal plane (i.e., abduction/adduction or internal/external rotation rather than flexion/extension) measurements [144,150]. Dahl et al. have validated an IMU system (Opal Gen 2) against a gold standard optical system for sports-related common movements (e.g., cutting, running, jumping) [151]. This study found a good level of agreement between the IMU and the optical system supporting the use of IMUs for sports-related movement/rehabilitation assessment, although the paper stated that continued validation and improvement of sensor accuracy is required. Jenkins et al. have investigated the use of an IMU (LSM6DSLTR) to monitor body configuration in relation to back injury risk in weightlifting exercises [152]. Taborri et al. have used IMU (Invensense ICM 20948) sensors integrated with a support vector machine algorithm to identify faults (e.g., illegal running) during race walking, which has the potential to assist officials in competitions (Figure 2b) [153]. The sensors used in these examples were not directly integrated into textiles, with textiles only used to hold the rigid enclosure of the IMUs in the form of a strap. In order to reduce the size of the sensing system and improve its flexibility, an IMU (MC6470) sensing filament has been developed by soldering the sensor on a narrow flexible circuit (<5 mm wide) and woven into textiles for monitoring joint angles and movement types (e.g., walking, running, climbing stairs) [154]. The flexible circuit was fabricated using photolithography and etching processes described in [13]. Four wires were soldered on the filament to implement the I2C protocol for data, clock, power and ground connections and woven into a textile that can worn on the arm to measure joint angle (Figure 2c). Although sports and fitness applications were not investigated in the paper, the technology is applicable to many sports and fitness applications involving the monitoring of joint movements. In addition to the standard IMU challenge of signal drift, which requires calibration and effective processing, integration of IMUs into a wearable item (e.g., sleeve, t-shirt, leggings) providing close and reliable contact with the skin is required to eliminate/reduce motion artefacts. Although the filament-sensing yarn approach has improved the integration within the textile, the establishment of scalable and cost-effective manufacturing is required to make it commercially variable.



С

Figure 2. (a) The assembled PCB prototypes with a 3D-printed case [152], reproduced with permission from the publisher Elsevier; (b) electrical circuit with a protective case [153]; (c) IMU filament embedded in a woven textile [154].

4.4. Electrodes for Heart Rate, Heart Rate Variability, and Muscle Excitation/Strength

ECG is the most common biosignal measured for the monitoring of cardiovascular health and to provide early detection of arrhythmias. The conventional Ag/AgCl hydrogel electrodes used in ECG monitoring are not suitable for long-term wearable applications due to the stickiness, moisture evaporation and ease of contamination of the electrodes. Textile-based dry electrodes have been developed for wearable ECG to overcome the disadvantages of hydrogel electrodes (Figure 3). The electrodes form an integral part of

the textile, which is then made into a wearable item that provides tight contact between the electrodes and the skin to reduce signal noise generated via the movement of the electrodes. Electrodes are mainly made by depositing conductive inks/paste on textiles using printing [155–157], coating the textile in the electrode material solution [158,159] or knitting [160,161], weaving [162,163], or embroidering [164,165] commercially available conductive yarns or bespoke conductive yarns such as graphene-based yarn and PDOT: PSS-modified yarn [166,167]. Sun et al. have developed an ECG t-shirt that can monitor an individual's heart rate during endurance training [168]. Ousaka et al. demonstrated the feasibility of continuously monitoring ECG during a full marathon to use the technology to prevent sudden cardiac accidents [169]. Bosco et al. investigated the use of a 12-lead ECG strapped to the waist to collect ECG data underwater to analyse heart rate, arrhythmias, conduction abnormalities and ischemic events in relation to various stages in diving [170].



Figure 3. (a) ECG bracelet made of printed electrodes with snap button connectors: inner layer (a-1), outer layer (a-2), bracelet form (a-3) [156], reproduced with permission from the publisher Elsevier; (b) electrodes made using the pad-dry-cure method and the design of an ECG bra [158], reproduced with permission from the publisher Elsevier.

EMG is used to monitor muscle excitation using the biological electrical activity signals of the muscle, which can provide useful indicators of muscle activation patterns, fatigue, and performance. EMG has an important role in analysing dynamic movement and understanding the role of a muscle in a specific movement [171]. The measurement of fatigue is important due to its negative impact on cognitive and motor performance, training effectiveness, and risk of injury [172]. The materials and manufacturing process of EMG electrodes are the same/similar to the ECG electrodes described above. Ohiri et al. have incorporated stretchable electrodes and interconnects within a set of athletic compression clothing, including a t-shirt, a pair of shorts, and calf sleeves for large-area EMG monitoring. The modular approach allows the monitoring of the biceps/triceps, quadriceps/hamstrings, and tibialis anterior/gastrocnemius muscles [173]. Liu et al. have developed a wearable EMG with two electrodes embedded in bicycle pants to monitor muscle fatigue during cycling. However, the study found that the EMG signal was affected by artificial noise due to the electrodes movement during exercise, leading to lower sensitivity compared to the Ag/AgCl electrodes [174].

4.5. Piezoresistive, Piezoelectrical, and Capacitive Materials for Pressure, Contact Force/Speed, and Respiratory Rate Measurement

Measuring the pressure/force applied to an object in sports and fitness activities (e.g., running, jumping, boxing) is another important parameter to monitor during training and to assess injury risk, e.g., from collisions. Piezoresistive, capacitive, and piezoelectric materials and devices are commonly used for monitoring pressure and force through the change in electrical resistance or capacitance or the generation of electrical charge in response to pressure or force, respectively. Functional materials are integrated into textiles using standard e-textile fabrication methods (e.g., printing, weaving, knitting, embroidery) to produce conformable and flexible smart garments that can be worn in sports/fitness

activities. For example, Ye et al. have developed an all-textile sensing system made of a three-dimensional spacer fabric dielectric layer sandwiched between two nickel-plated plain woven fabric electrodes to detect pressure, distance, and speed and which exhibits good potential for real-time monitoring of human physical combat sports (Figure 4a) [175]. Ma et al. have developed a dual tactile-tension sensing textile made of a three-dimensional spacer fabric sandwiched between two layers of woven textile electrodes (Figure 4b). The textile electrodes are made of silver yarns and core-sheath conductive yarns that can monitor the change of the impendence and capacitance to detect stretching and pressure in taekwondo training [176].



Figure 4. All textile-based sensors to measure (**a**) touchless and tactile signals in boxing: schematic diagram (a-1), a sensor array sewn into a garment (a-2), capacitance response for slow, medium and quick punches (a-3), local amplification diagram response to slow, medium and quick punches (a-4), comparison of punching speed (a-5) and punching force (a-6) [175], reproduced with permission from the publisher Elsevier; (**b**) stretching and pressure in taekwondo [176], reproduced with permission from the publisher Elsevier.

These sensors can also be used to monitor the respiratory rate by detecting the expansion and contraction of the chest during inhalation and exhalation, which can be indicated by the change of resistance and capacitance. Sensors can be embedded in chest straps or smart garments to allow real-time monitoring, such as piezoresistive sensors made of silver-plated knitted textiles for respiratory rate monitoring [177].

Other applications include temperature-sensing textiles to monitor overheating or hypothermia, moisture-sensing textiles to monitor sweat levels and hydration status, SpO2

sensing textiles to monitor respiratory and cardiovascular health, and electroencephalography (EEG) sensing textiles to monitor cognitive status [147,178].

5. Challenges and Future Opportunities

There are certain technical limitations of e-textile implementations that constrain the practical application and wide adoption of smart fabric technologies. The following challenges need to be considered in the research and development of e-textiles for practical applications:

- The device's sensing accuracy and the influence of the textile or clothing on the accuracy are challenges in many applications and will require calibration and correct fitting of the wearables. For example, the data gathered from inertial sensors on garments will depend upon how tight fitting the clothing is, as well as potential sensor-to-segment alignment algorithms/procedures. The accuracy can also be affected by sampling rate, sensor drift, and the external environment (e.g., temperature, magnetic interference). Poor ECG/EMG electrode-skin contact will increase the electrode-skin impedance and noise generated by motion artefacts. Accuracy can also be affected by the change in humidity (e.g., sweating), an unstable connection between the electrodes and the electronics, and the degradation of the electrodes. Many piezoresistive materials made of carbon particle-filled polymers exhibit non-monotonic strain response, especially under dynamic conditions due to the cyclical displacement of the conductive network, which can be affected by strain rate, strain history, and maximum applied strain [179];
- The connector used in e-textiles is another component that impacts the size/flexibility and reliability/durability of the sensing system. Sensors must be reliably connected to signal-processing electronics and power sources, and unobtrusively embedding these technologies is not straightforward. Connectors used in e-textiles that allow detachment of the electronic/sensors (e.g., snap fastener, pogo pins, magnetic connector) introduce rigid components with considerable size compared to the sensors, and those providing permanent attachment create potential failure points due to the poor adhesion (e.g., conductive adhesive) and limited capability to accommodate the level of flexing/bending of the textile causing stress on the joints (e.g., soldering);
- Electronic and sensing technologies incorporated during the manufacture of the textile must survive the associated manufacturing process (e.g., weaving, knitting, surface finishing), and e-textile processes must be compatible with mass manufacturing to allow scale-up and reduction of cost. During use, textiles routinely experience physical conditions (e.g., physical wear, bending and flexing, exposure to liquids, and washing) and ensuring solutions are robust and reliable is a significant challenge;
- E-textiles are reliant on conventional primary or secondary batteries, and these are bulky, rigid, and incompatible with the feel of the textile. Although the sensor components are very small (e.g., IMUs) and flexible (e.g., electrodes for biosignals), the presence of the battery has significantly impacted the size and flexibility of the overall system. The integration of the battery with the textile is poor, and the battery will need to be removed prior to washing, which happens regularly because of exercise. The inherently rigid and bulky nature of the battery is intrusive and can spoil the user's perception of the e-textile technology;
- Co-design with target users to ensure their requirement is considered from the outset of the project. The design needs to consider different requirements associated with sex/gender, age, body size/shape, physical mobility, and digital literacy. Testing with a large user cohort is essential to improve the data quality and explore/address the challenges that may not be noticeable within a small user group;
- E-textiles-based applications that involve data collection, transmission, storage and sharing must consider safety and the implications of data management and associated ethics/privacy issues. The ownership of the data and how it will be used in a safe and ethical way needs to be considered;

- The lack of standards and legislation. The e-textile standards are under development and have not been widely tested by technology/product developers. This has led to the lack of standardised processes and methods, creating challenges in assessing and comparing different work/products. Legislation and incentives are required to address the environmental issues posed by the e-textile;
- Sustainability and circularity need to be considered from the design stage and across
 the whole product development and product life cycle to minimise the impact on the
 environment. Products need to be robust to allow longevity. Disassembly needs to
 be considered to separate different components used in the e-textile for reuse and
 recycling. In addition to the technology/product developers, other stakeholders
 (e.g., consumers) need to become involved and accessible infrastructures need to be
 established to achieve reliable and effective circularity.

Research is required to address the challenges above, which creates opportunities for advancing e-textile technologies and developing fit-for-purpose products. Continuous development of new materials/components and advanced manufacturing processes are required to improve the sensitivity, reliability, durability, and level of integration of textilebased technologies, enabling their long-term use in sports and fitness. Interacting etextiles with other technologies (e.g., digital technologies, AI) will add new functions for e-textile-based wearable technologies. For example, using AI tools built on the analysis of substantial movement and performance data collected by the e-textile sensors will allow the development of personalised training plans and prediction of performance and rehabilitation progress. Co-design with end users and testing prototypes with a diverse group representing various characteristics of the intended users will allow the researcher to capture valuable insights and feedback to improve the design and increase the likelihood of product market fit. Responsible innovation can only be achieved and implemented with the contribution of different stakeholders involved (e.g., technology developers, manufacturers, distributors, end users, and policymakers), enabling e-textiles to become user-centric, cost-effective, and sustainable solutions.

6. Conclusions

Wearable technologies have been widely used in sports and fitness applications at both recreational and elite levels to track performance and health conditions. Wearable sensors provide useful data in vital biosignals and key performance parameters (e.g., speed, distance, acceleration) that can inform training to improve efficacy while reducing the risk of injury. E-textiles provide a platform for the ubiquitous deployment of wearable technologies due to the soft and comfortable nature of the textiles and their suitability for everyday wearing. The level of integration varies among different techniques and applications, from directly inserting the modular sensor/electronic unit (e.g., IMUs, GPS) into a pocket of the garment to having the sensing material truly embedded into the textile to make a sensing textile (e.g., ECG/EMG electrodes, force/pressure sensing fabric) that are connected to a detachable electronic unit. These two types of integration are the most common methods used in commercial e-textile products, and the fully integrated e-textiles system (e.g., electronic units, including batteries) is still in its development stage.

Further advancement is required in order to gain the full benefit of e-textile-based sensors. This includes the development of new materials and components with improved accuracy, reliability and durability; integration of the e-textiles system with miniaturised components to reduce the overall size and its impact on the sports/fitness garment; manufacturing processes that are scalable and cost-effective for both prototyping and volume production; co-design and with end users to ensure product market fit, standardised protocols in data management/privacy and performance testing. In addition, sustainability and circularity need to be considered in each stage of development, as well as the life and post-life cycles of the products. The range of disciplines involved in e-textile research should be broadened to identify new opportunities in sports/fitness applications and ad-

dress the challenges directly related to the e-textiles and these related to broader disciplines (e.g., energy efficiency, data-informed decisions).

Author Contributions: B.I. Collected the information from the Sports and Fitness Applications workshop. S.A.M.-N. and A.C. authored the sports technology aspects. S.B. and K.Y. reviewed the existing wearable technologies and e-textiles technologies for sports and fitness, analysed the drivers and constraints, identified the future challenges, and wrote these parts of the paper. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the EPSRC in the UK under grant number EP/R031738/1. The work of Steve Beeby was supported by the Royal Academy of Engineering under the Chairs in Emerging Technologies Scheme (grant number: CiET2021\89).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data are contained within the article.

Acknowledgments: This review was built on discussions at the "E-Textiles for Sports and Fitness Applications" workshop organized by the E-textiles Network funded by EPSRC. The authors would like to thank Peter Lazou, Naomi McGregor, and Martin Ashby for presenting at the workshop. The authors would like to thank all the workshop participants.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- 1. Rum, L.; Sten, O.; Vendrame, E.; Belluscio, V.; Camomilla, V.; Vannozzi, G.; Truppa, L.; Notarantonio, M.; Sciarra, T.; Lazich, A.; et al. Wearable Sensors in Sports for Persons with Disability: A Systematic Review. *Sensors* **2021**, *21*, 1858. [CrossRef]
- 2. Aroganam, G.; Manivannan, N.; Harrison, D. Review on Wearable Technology Sensors Used in Consumer Sport Applications. Sensors 2019, 19, 1983. [CrossRef]
- 3. Clermont, C.A.; Duffett-Leger, L.; Hettinga, B.A.; Ferber, R. Runners' Perspectives on 'Smart' Wearable Technology and Its Use for Preventing Injury. *Int. J. Hum. Comput. Interact.* **2020**, *36*, 31–40. [CrossRef]
- 4. Benson, L.C.; Räisänen, A.M.; Volkova, V.G.; Pasanen, K.; Emery, C.A. Workload a-WEAR-ness: Monitoring Workload in Team Sports With Wearable Technology. A Scoping Review. J. Orthop. Sports Phys. Ther. 2020, 50, 549–563. [CrossRef]
- Di Paolo, S.; Lopomo, N.F.; Della Villa, F.; Paolini, G.; Figari, G.; Bragonzoni, L.; Grassi, A.; Zaffagnini, S. Rehabilitation and Return to Sport Assessment after Anterior Cruciate Ligament Injury: Quantifying Joint Kinematics during Complex High-Speed Tasks through Wearable Sensors. *Sensors* 2021, 21, 2331. [CrossRef]
- Rana, M.; Mittal, V. Wearable Sensors for Real-Time Kinematics Analysis in Sports: A Review. IEEE Sens. J. 2021, 21, 1187–1207. [CrossRef]
- Rupp, M.A.; Michaelis, J.R.; McConnell, D.S.; Smither, J.A. The Role of Individual Differences on Perceptions of Wearable Fitness Device Trust, Usability, and Motivational Impact. *Appl. Ergon.* 2018, 70, 77–87. [CrossRef]
- 8. Komolafe, A.; Zaghari, B.; Torah, R.; Weddell, A.S.; Khanbareh, H.; Tsikriteas, Z.M.; Vousden, M.; Wagih, M.; Jurado, U.T.; Shi, J.; et al. E-Textile Technology Review–From Materials to Application. *IEEE Access* **2021**, *9*, 97152–97179. [CrossRef]
- 9. Facts & Factors. *Global E-Textiles and Smart Clothing Market Size, Share;* Facts & Factors: Pune, India, 2021.
- 10. STATSports. Available online: https://statsports.com/apex-athlete-series (accessed on 9 December 2023).
- 11. Catapult Sports. Available online: https://one.catapultsports.com/gb/ (accessed on 9 December 2023).
- 12. Shishoo, R. (Ed.) *Textiles for Sportswear*, 1st ed.; Elsevier: Amsterdam, The Netherlands, 2015.
- 13. Komolafe, A.O.; Torah, R.N.; Wei, Y.; Nunes-Matos, H.; Li, M.; Hardy, D.; Dias, T.; Tudor, M.J.; Beeby, S.P. Integrating Flexible Filament Circuits for E-textile Applications. *Adv. Mater. Technol.* **2019**, *4*, 1900176. [CrossRef]
- 14. Wicaksono, I.; Tucker, C.I.; Sun, T.; Guerrero, C.A.; Liu, C.; Woo, W.M.; Pence, E.J.; Dagdeviren, C. A Tailored, Electronic Textile Conformable Suit for Large-Scale Spatiotemporal Physiological Sensing in vivo. *NPJ Flex. Electron.* **2020**, *4*, 5. [CrossRef]
- 15. Prevayl. Available online: https://www.prevayl.com/ (accessed on 9 December 2023).
- 16. Success Stories—Learn about How Equivital Has Made a Difference. Available online: https://equivital.com/success-stories (accessed on 20 January 2024).
- 17. Derrick, T.R. The Need for Interdisciplinary Research in Sports Biomechanics. Sports Biomech. 2004, 3, 1–4.
- 18. McCrory, P.F. Interdisciplinary Research in Sports Medicine. Br. J. Sports Med. 2006, 40, 259–260.
- 19. Stone, M.S. A Call for Interdisciplinary Research in Sports Science. J. Strength. Cond. Res. 2008, 22, 3–4.
- 20. Timmons, B.W.; Steadward, R.D. The Need for Interdisciplinary Research in Sports Science. *Appl. Physiol. Nutr. Metab.* **2010**, 35, 1–3.

- 21. Lemmink, K.A.P.M. Interdisciplinary Approaches in Sports Science Research: From Theory to Practice. *Eur. J. Sport Sci.* 2015, 15, 181–182.
- Delecroix, M.; McCall, A.; Dawson, B.; Winwood, P. Interdisciplinary Approach to the Design of Effective Resistance Training in Team Sports. Strength. Cond. J. 2018, 40, 17–24.
- 23. Fagher, K.; Forsberg, A.; Jacobsson, J.; Timpka, T. Interdisciplinary Research is Needed to Improve Injury Prevention and Rehabilitation in Sports. *J. Orthop. Sports Phys. Ther.* **2019**, *49*, 476–478.
- 24. Hennessey, T.; Kilty, J. Embracing Complexity: Interdisciplinary Perspectives in Sports Science Research. J. Sports Sci. 2020, 38, 2073–2074.
- Lupo, C.; Ungureanu, A.N.; Varalda, M.; Cortis, C. Editorial: Advances in Interdisciplinary Approaches in Sports Science Research. Front. Psychol. 2020, 11, 2096.
- Schwesig, R.; Struder, H.K.; Faude, O. Need for Interdisciplinary Sports Science Research in the era of COVID-19. *Eur. J. Sport Sci.* 2021, 21, 248–251.
- 27. Reid, C.; Stewart, E.; Thorne, G. Multidisciplinary Sport Science Teams in Elite Sport: Comprehensive Servicing or Conflict and Confusion? *Sport Psychol.* 2004, *18*, 204–217. [CrossRef]
- 28. Goebert, C. Augmented reality in sport marketing: Uses and directions. Sports Innov. J. 2020, 1, 134–151. [CrossRef]
- 29. Nguyen, V.; Bin, Y.S.; Campbell, A. The Application of Virtual Reality in Sports: A Systematic Review. *Front. Psychol.* **2021**, *11*, 620510.
- 30. Neumann, D.L.; Moffitt, R.L.; Thomas, P.R.; Loveday, K.; Watling, D.P.; Lombard, C.L.; Tremeer, M.A. A Systematic Review of the Application of Interactive Virtual Reality to Sport. *Virtual Real.* **2018**, *22*, 183–198. [CrossRef]
- 31. Farley, O.R.; Spencer, K.; Baudinet, L. Virtual Reality in Sports Coaching, skill Acquisition and Application to Surfing: A Review. *J. Hum. Sport Exerc.* **2020**, *15*, 535–548. [CrossRef]
- Pitsch, W.; Avenarius, H. Inequality in sports technology: A Critical Reflection on the Role of Technological Artifacts in Sport. Sport Ethics Philos. 2019, 13, 270–284.
- Ferreira, L.; Dias, G.; Almeida, F. Elite sports and technology: The Reproduction of Inequality. Int. J. Sociol. Soc. Policy 2020, 40, 474–488.
- 34. Clayton, K.; Shulman, J. The Politics of Technology and the Future of Sports. Commun. Sport 2020, 8, 651–664.
- 35. Rosenbaum, S.; Hobson-Powell, A.; Davison, K.; Stanton, R.; Craft, L.L.; Duncan, M.; Ward, P.B. The Role of Sport, Exercise, and Physical Activity in Closing the Life Expectancy Gap for People with Mental Illness: An International Consensus Statement by Exercise and Sports Science Australia, American College of Sports Medicine, British Association of Sport and Exercise Science, and Sport and Exercise Science New Zealand. *Transl. J. Am. Coll. Sports Med.* **2018**, *3*, 72–73.
- Banister, E.W.; Calvert, T.W.; Savage, M.V.; Bach, T.M. A Systems Model of Training for Athletic Performance. Aust. J. Sports Med. 1975, 7, 57–61.
- Wang, T.; Lin, Z.; Day, R.E.; Gardiner, B.; Landao-Bassonga, E.; Rubenson, J.; Kirk, T.B.; Smith, D.W.; Lloyd, D.G.; Hardisty, G.; et al. Programmable Mechanical Stimulation Influences Tendon Homeostasis in a Bioreactor System. *Biotechnol. Bioeng.* 2012, 110, 1495–1507. [CrossRef]
- 38. Bompa, T.O.; Buzzichelli, C. Periodization Training for Sports, 3rd ed.; Human Kinetics: Champaign, IL, USA, 2015.
- Calvert, T.W.; Banister, E.W.; Savage, M.V.; Bach, T. A Systems Model of the Effects of Training on Physical Performance. *IEEE Trans. Syst. Man. Cybern.* 1976, SMC-6, 94–102. [CrossRef]
- 40. Halson, S.L. Monitoring Training Load to Understand Fatigue in Athletes. Sports Med. 2014, 44 (Suppl. S2), 139–147. [CrossRef]
- 41. Impellizzeri, F.M.; Rampinini, E.; Coutts, A.J.; Sassi, A.; Marcora, S.M. Use of RPE-Based Training Load in Soccer. *Med. Sci. Sports Exerc.* **2004**, *36*, 1042–1047. [CrossRef]
- Impellizzeri, F.M.; Marcora, S.M.; Coutts, A.J. Internal and External Training Load: 15 Years On. Int. J. Sports Physiol. Perform. 2019, 14, 270–273. [CrossRef]
- 43. Impellizzeri, F.M.; Rampinini, E.; Marcora, S.M. Physiological Assessment of Aerobic Training in Soccer. J. Sports Sci. 2005, 23, 583–592. [CrossRef]
- 44. Impellizzeri, F.M.; Jeffries, A.C.; Weisman, A.; Coutts, A.C.; McCall, A.; McLaren, S.J.; Kalkhoven, J.T. A Clarification of the (Mis)use of the Term 'Load' in Sport and Exercise Science: Why it is Appropriate and Scientific. *SportRxiv* 2021. [CrossRef]
- 45. Impellizzeri, F.M.; Shrier, I.; McLaren, S.J.; Coutts, A.J.; McCall, A.; Slattery, K.; Jeffries, A.C.; Kalkhoven, J.T. Understanding Training Load as Exposure and Dose. *Sports Med.* **2023**, *53*, 1667–1679. [CrossRef]
- 46. Vanrenterghem, J.; Nedergaard, N.J.; Robinson, M.A.; Drust, B. Training Load Monitoring in Team Sports: A Novel Framework Separating Physiological and Biomechanical Load-Adaptation Pathways. *Sports Med.* **2017**, *47*, 2135–2142. [CrossRef]
- Myers, J.; Kaminsky, L.A.; Lima, R.; Christle, J.W.; Ashley, E.; Arena, R. A reference equation for normal standards for VO₂ max: Analysis from the Fitness Registry and the Importance of Exercise National Database (FRIEND Registry). *Prog. Cardiovasc. Dis.* 2017, 60, 21–29. [CrossRef]
- 48. Bradshaw, D.I.; George, J.D.; Hyde, A.; LaMonte, M.J.; Vehrs, P.R.; Hager, R.L.; Yanowitz, F.G. An accurate VO₂ max nonexercise regression model for 18–65-year-old adults. *Res. Q. Exerc. Sport* **2005**, *76*, 426–432. [CrossRef]
- 49. Ruggeri, E.; Matzeu, G.; Vergine, A.; De Nicolao, G.; Omenetto, F.G. Paper-Based Wearable Patches for Real-Time, Quantitative Lactate Monitoring. *Adv. Sens. Res.* 2023, 2300141. [CrossRef]

- 50. Verheul, J.; Nedergaard, N.J.; Vanrenterghem, J.; Robinson, M.A. Measuring Biomechanical Loads in Team Sports—From Lab to Field. *Sci. Med. Footb.* 2020, *4*, 246–252. [CrossRef]
- Malone, J.J.; Lovell, R.; Varley, M.C.; Coutts, A.J. Unpacking the Black Box: Applications and Considerations for Using GPS Devices in Sport. Int. J. Sports Physiol. Perform. 2017, 12, S2-18–S2-26. [CrossRef]
- 52. Jackson, B.M.; Polglaze, T.; Dawson, B.; King, T.; Peeling, P. Comparing Global Positioning System and Global Navigation Satellite System Measures of Team-Sport Movements. *Int. J. Sports Physiol. Perform.* **2018**, *13*, 1005–1010. [CrossRef]
- Connor, M.; Beato, M.; O'Neill, M. Adaptive Athlete Training Plan Generation: An Intelligent Control Systems Approach. J. Sci. Med. Sport 2022, 25, 351–355. [CrossRef]
- 54. Serpiello, F.R.; Hopkins, W.G.; Barnes, S.; Tavrou, J.; Duthie, G.M.; Aughey, R.J.; Ball, K. Validity of an Ultra-Wideband Local Positioning System to Measure Locomotion in Indoor Sports. *J. Sports Sci.* **2018**, *36*, 1727–1733. [CrossRef]
- Pino-Ortega, J.; Oliva-Lozano, J.M.; Gantois, P.; Nakamura, F.Y.; Rico-González, M. Comparison of the validity and reliability of local positioning systems against other tracking technologies in team sport: A systematic review. *Proc. Inst. Mech. Eng. Part. P J. Sports Eng. Technol.* 2022, 236, 73–82. [CrossRef]
- 56. Barrett, S.; Midgley, A.; Lovell, R. PlayerLoad[™]: Reliability, Convergent Validity, and Influence of Unit Position during Treadmill Running. *Int. J. Sports Physiol. Perform.* **2014**, *9*, 945–952. [CrossRef]
- 57. Nicolella, D.P.; Torres-Ronda, L.; Saylor, K.J.; Schelling, X. Validity and reliability of an accelerometer-based player tracking device. *PLoS ONE* **2018**, *13*, e0191823. [CrossRef]
- 58. Beato, M.; De Keijzer, K.L.; Carty, B.; Connor, M. Monitoring Fatigue During Intermittent Exercise with Accelerometer-Derived Metrics. *Front. Physiol.* **2019**, *10*, 780. [CrossRef]
- 59. Young, D.; Coratella, G. Acceleration, Deceleration and Dynamic Stress Load in Elite Hurling: A Between-Quarter and Between-Position Comparison. *Sports* **2021**, *9*, 10. [CrossRef]
- 60. McErlain-Naylor, S.A.; King, M.A.; Allen, S.J. Surface acceleration transmission during drop landings in humans. J. Biomech. 2021, 118, 110269. [CrossRef]
- 61. Shorten, M.R.; Winslow, D.S. Spectral Analysis of Impact Shock during Running. Int. J. Sport Biomech. 1992, 8, 288–304. [CrossRef]
- 62. Matijevich, E.S.; Branscombe, L.M.; Scott, L.R.; Zelik, K.E. Ground reaction force metrics are not strongly correlated with tibial bone load when running across speeds and slopes: Implications for science, sport and wearable tech. *PLoS ONE* **2019**, *14*, e0210000. [CrossRef]
- 63. Sparks, M.; Coetzee, B.; Gabbett, T.J. Internal and External Match Loads of University-Level Soccer Players: A Comparison Between Methods. *J. Strength Cond. Res.* 2017, *31*, 1072–1077. [CrossRef]
- 64. McLean, B.D.; Cummins, C.; Conlan, G.; Duthie, G.; Coutts, A.J. The Fit Matters: Influence of Accelerometer Fitting and Training Drill Demands on Load Measures in Rugby League Players. *Int. J. Sports Physiol. Perform.* **2018**, *13*, 1083–1089. [CrossRef]
- 65. Mannini, A.; Sabatini, A.M. Machine Learning Methods for Classifying Human Physical Activity from On-Body Accelerometers. Sensors 2010, 10, 1154–1175. [CrossRef]
- Chambers, R.; Gabbett, T.J.; Cole, M.H.; Beard, A. The Use of Wearable Microsensors to Quantify Sport-Specific Movements. Sports Med. 2015, 45, 1065–1081. [CrossRef]
- 67. Ghosh, I.; Ramamurthy, S.R.; Chakma, A.; Roy, N. DeCoach: Deep Learning-based Coaching for Badminton Player Assessment. *Pervasive Mob. Comput.* **2022**, *83*, 101608. [CrossRef]
- 68. Perri, T.; Reid, M.; Murphy, A.; Howle, K.; Duffield, R. Validating an algorithm from a trunk-mounted wearable sensor for detecting stroke events in tennis. *J. Sports Sci.* 2022, *40*, 1168–1174. [CrossRef]
- 69. Beaupré, G.S.; Orr, T.E.; Carter, D.R. An approach for time-dependent bone modeling and remodeling—Application: A preliminary remodeling simulation. *J. Orthop. Res.* **1990**, *8*, 662–670. [CrossRef]
- Ahola, R.; Korpelainen, R.; Vainionpää, A.; Jämsä, T. Daily impact score in long-term acceleration measurements of exercise. J. Biomech. 2010, 43, 1960–1964. [CrossRef]
- Burland, J.P.; Outerleys, J.B.; Lattermann, C.; Davis, I.S. Reliability of wearable sensors to assess impact metrics during sportspecific tasks. J. Sports Sci. 2021, 39, 406–411. [CrossRef]
- 72. Armitage, M.; Beato, M.; McErlain-Naylor, S.A. Inter-unit reliability of IMU Step metrics using IMeasureU Blue Trident inertial measurement units for running-based team sport tasks. *J. Sports Sci.* 2021, 39, 1512–1518. [CrossRef]
- 73. Zandbergen, M.A.; Ter Wengel, X.J.; van Middelaar, R.P.; Buurke, J.H.; Veltink, P.H.; Reenalda, J. Peak tibial acceleration should not be used as indicator of tibial bone loading during running. *Sports Biomech.* **2023**, 1–18. [CrossRef]
- 74. Matijevich, E.S.; Scott, L.R.; Volgyesi, P.; Derry, K.H.; Zelik, K.E. Combining wearable sensor signals, machine learning and biomechanics to estimate tibial bone force and damage during running. *Hum. Mov. Sci.* **2020**, *74*, 102690. [CrossRef]
- 75. Elstub, L.J.; Nurse, C.A.; Grohowski, L.M.; Volgyesi, P.; Wolf, D.N.; Zelik, K.E. Tibial bone forces can be monitored using shoe-worn wearable sensors during running. *J. Sports Sci.* **2022**, *40*, 1741–1749. [CrossRef]
- 76. Slade, P.; Habib, A.; Hicks, J.L.; Delp, S.L. An open-source and wearable system for measuring 3D human motion in real-time. *IEEE Trans. Biomed. Eng.* **2022**, *69*, 678–688. [CrossRef]
- Al Borno, M.; O'Day, J.; Ibarra, V.; Dunne, J.; Seth, A.; Habib, A.; Ong, C.; Hicks, J.; Uhlrich, S.; Delp, S. OpenSense: An open-source toolbox for inertial-measurement-unit-based measurement of lower extremity kinematics over long durations. *J. NeuroEngineering Rehabil.* 2022, 19, 22. [CrossRef]

- 78. Gholami, M.; Napier, C.; Menon, C. Estimating Lower Extremity Running Gait Kinematics with a Single Accelerometer: A Deep Learning Approach. *Sensors* 2020, 20, 2939. [CrossRef]
- 79. Lim, H.; Kim, B.; Park, S. Prediction of Lower Limb Kinetics and Kinematics during Walking by a Single IMU on the Lower Back Using Machine Learning. *Sensors* 2020, 20, 130. [CrossRef]
- Liew, B.X.W.; Rügamer, D.; Zhai, X.; Wang, Y.; Morris, S.; Netto, K. Comparing shallow, deep, and transfer learning in predicting joint moments in running. J. Biomech. 2021, 129, 110820. [CrossRef]
- 81. Mombaur, K.; Truong, A.; Laumond, J.P. From human to humanoid locomotion—An inverse optimal control approach. *Auton. Robot.* **2010**, *28*, 369–383. [CrossRef]
- Mainprice, J.; Hayne, R.; Berenson, D. Predicting human reaching motion in collaborative tasks using Inverse Optimal Control and iterative re-planning. In Proceedings of the 2015 IEEE International Conference on Robotics and Automation (ICRA), Seattle, WA, USA, 26–30 May 2015; pp. 885–892.
- 83. Dorschky, E.; Nitschke, M.; Seifer, A.K.; van den Bogert, A.J.; Eskofier, B.M. Estimation of gait kinematics and kinetics from inertial sensor data using optimal control of musculoskeletal models. *J. Biomech.* **2019**, *95*, 109278. [CrossRef]
- Matijevich, E.S.; Volgyesi, P.; Zelik, K.E. A Promising Wearable Solution for the Practical and Accurate Monitoring of Low Back Loading in Manual Material Handling. *Sensors* 2021, 21, 340. [CrossRef]
- 85. Ross, G.B.; Dowling, B.; Troje, N.F.; Fischer, S.L.; Graham, R.B. Classifying Elite From Novice Athletes Using Simulated Wearable Sensor Data. *Front. Bioeng. Biotechnol.* 2020, *8*, 814. [CrossRef]
- Warmenhoven, J.; Harrison, A.; Quintana, D.S.; Hooker, G.; Gunning, E.; Bargary, N. Unlocking Sports Medicine Research Data While Maintaining Participant Privacy via Synthetic Datasets. Available online: https://osf.io/preprints/sportrxiv/f3rz7 (accessed on 9 December 2023).
- 87. Azizi, Z.; Zheng, C.; Mosquera, L.; Pilote, L.; El Emam, K. Can synthetic data be a proxy for real clinical trial data? A validation study. *BMJ Open* **2021**, *11*, e043497. [CrossRef]
- 88. Bicer, M.; Phillips, A.T.M.; Melis, A.; McGregor, A.H.; Modenese, L. Generative deep learning applied to biomechanics: A new augmentation technique for motion capture datasets. *J. Biomech.* **2022**, *144*, 111301. [CrossRef]
- 89. Qureshi, S.M.; Purdy, N.; Greig, M.A.; Kelly, H.; vanDeursen, A.; Neumann, W.P. Developing a simulation tool to quantify biomechanical load and quality of care in nursing. *Ergonomics* **2022**, *66*, 886–890. [CrossRef]
- 90. Demangeot, Y.; Whiteley, R.; Gremeaux, V.; Degache, F. The load borne by the Achilles tendon during exercise: A systematic review of normative values. *Scand. J. Med. Sci. Sports* **2023**, *33*, 110–126. [CrossRef]
- Sancho, I.; Willy, R.W.; Morrissey, D.; Malliaras, P.; Lascurain-Aguirrebeña, I. Achilles tendon forces and pain during common rehabilitation exercises in male runners with Achilles tendinopathy. A laboratory study. *Phys. Ther. Sport* 2023, 60, 26–33. [CrossRef]
- 92. DeJong Lempke, A.F.; Stephens, S.L.; Fish, P.N.; Thompson, X.D.; Hart, J.M.; Hryvniak, D.J.; Rodu, J.S.; Hertel, J. Sensor-based gait training to reduce contact time for runners with exercise-related lower leg pain: A randomised controlled trial. *BMJ Open Sport Exerc. Med.* **2022**, *8*, e001293. [CrossRef]
- 93. Galofaro, E.; D'Antonio, E.; Lotti, N.; Masia, L. Rendering Immersive Haptic Force Feedback via Neuromuscular Electrical Stimulation. *Sensors* 2022, 22, 5069. [CrossRef]
- 94. Te Brake, M.; Stolwijk, N.; Staal, B.; Van Hooren, B. Using beat frequency in music to adjust running cadence in recreational runners: A randomized multiple baseline design. *Eur. J. Sport Sci.* **2023**, *23*, 345–354. [CrossRef]
- 95. Van den Berghe, P.; Derie, R.; Bauwens, P.; Gerlo, J.; Segers, V.; Leman, M.; De Clercq, D. Reducing the peak tibial acceleration of running by music-based biofeedback: A quasi-randomized controlled trial. *Scand. J. Med. Sci. Sports* 2022, 32, 698–709. [CrossRef]
- 96. Hughes, G.T.G.; Camomilla, V.; Vanwanseele, B.; Harrison, A.J.; Fong, D.T.P.; Bradshaw, E.J. Novel technology in sports biomechanics: Some words of caution. *Sports Biomech.* **2021**, 1–9. [CrossRef]
- 97. Preatoni, E.; Bergamini, E.; Fantozzi, S.; Giraud, L.I.; Orejel Bustos, A.S.; Vannozzi, G.; Camomilla, V. The Use of Wearable Sensors for Preventing, Assessing, and Informing Recovery from Sport-Related Musculoskeletal Injuries: A Systematic Scoping Review. *Sensors* 2022, 22, 3225. [CrossRef]
- 98. Wylde, M.; Adilah Masismadi, N.; Chee Yong, L.; Callaway, A.; Williams, J.M. Placement of inertial measurement units in Racket Sports: Perceptions of coaches for IMU use during training and competition. *Int. J. Racket Sports Sci.* 2021, *3*, 45–55. [CrossRef]
- 99. Lloyd, D. The future of in-field sports biomechanics: Wearables plus modelling compute real-time in vivo tissue loading to prevent and repair musculoskeletal injuries. *Sports Biomech.* **2021**, 1–29. [CrossRef]
- 100. Santos-Gago, J.M.; Ramos-Merino, M.; Vallarades-Rodriguez, S.; Álvarez-Sabucedo, L.M.; Fernández-Iglesias, M.J.; García-Soidán, J.L. Innovative Use of Wrist-Worn Wearable Devices in the Sports Domain: A Systematic Review. *Electronics* 2019, 8, 1257. [CrossRef]
- Cosoli, G.; Spinsante, S.; Scalise, L. Wrist-worn and chest-strap wearable devices: Systematic review on accuracy and metrological characteristics. *Measurement* 2020, 159, 107789. [CrossRef]
- Passos, J.; Lopes, S.I.; Clemente, F.M.; Moreira, P.M.; Rico-González, M.; Bezerra, P.; Rodrigues, L.P. Wearables and Internet of Things (IoT) Technologies for Fitness Assessment: A Systematic Review. *Sensors* 2021, 21, 5418. [CrossRef]
- Bent, B.; Goldstein, B.A.; Kibbe, W.A.; Dunn, J.P. Investigating sources of inaccuracy in wearable optical heart rate sensors. NPJ Digit. Med. 2020, 3, 18. [CrossRef]

- 104. Cosoli, G.; Antognoli, L.; Veroli, V.; Scalise, L. Accuracy and Precision of Wearable Devices for Real-Time Monitoring of Swimming Athletes. *Sensors* 2022, 22, 4726. [CrossRef]
- 105. Lopez, G.; Abe, S.; Hashimoto, K.; Yokokubo, A. On-Site Personal Sport Skill Improvement Support Using Only a Smartwatch. In Proceedings of the 2019 IEEE International Conference on Pervasive Computing and Communications Workshops (PerCom Workshops), Kyoto, Japan, 11–15 March 2019; pp. 158–164.
- 106. Whoop. Available online: https://www.whoop.com/ (accessed on 9 December 2023).
- 107. Beast Technologies. Available online: https://thisisbeast.com/blogs/solutions/beastsensor (accessed on 9 December 2023).
- Thompson, S.W.; Rogerson, D.; Dorrell, H.F.; Ruddock, A.; Barnes, A. The Reliability and Validity of Current Technologies for Measuring Barbell Velocity in the Free-Weight Back Squat and Power Clean. Sports 2020, 8, 94. [CrossRef]
- 109. Weippert, M.; Kumar, M.; Kreuzfeld, S.; Arndt, D.; Rieger, A.; Stoll, R. Comparison of three mobile devices for measuring R-R intervals and heart rate variability: Polar S810i Suunto t6 and an ambulatory ECG system. *Eur. J. Appl. Physiol.* 2010, 109, 779–786. [CrossRef]
- 110. Gilgen-Ammann, R.; Schweizer, T.; Wyss, T. RR interval signal quality of a heart rate monitor and an ECG Holter at rest and during exercise. *Eur. J. Appl. Physiol.* **2019**, *119*, 1525–1532. [CrossRef]
- 111. Camomilla, V.; Bergamini, E.; Fantozzi, S.; Vannozzi, G. Trends supporting the in-field use of wearable inertial sensors for sport performance evaluation: A systematic review. *Sensors* 2018, *18*, 873. [CrossRef]
- 112. Pla, R.; Ledanois, T.; Simbana, E.D.; Aubry, A.; Tranchard, B.; Toussaint, J.-F.; Sedeaud, A.; Seifert, L. Spatial-temporal variables for swimming coaches: A comparison study between video and TritonWear sensor. *Int. J. Sports Sci. Coach.* 2021, 16, 1271–1280. [CrossRef]
- 113. Movella Inc. Available online: https://www.xsens.com/products/mtw-awinda (accessed on 9 December 2023).
- 114. Held, J.P.O.; Yu, K.; Pyles, C.; Veerbeek, J.M.; Bork, F.; Heining, S.M.; Navab, N.; Luft, A.R. Augmented reality–based rehabilitation of gait impairments: Case report. *JMIR Mhealth Uhealth* **2020**, *8*, e17804. [CrossRef]
- Marreiros, S.; Schepers, M.; Bellusci, G.; de Zee, M.; Andersen, M.S. Comparing performance of two methods to process inertial data in gait analysis. In Proceedings of the XXVI Congress of the International Society of Biomechanics, Brisbane, Australia, 23–27 July 2017.
- 116. Wouda, F.J.; Giuberti, M.; Bellusci, G.; Maartens, E.; Reenalda, J.; Van Beijnum, B.J.F.; Veltink, P.H. On the validity of different motion capture technologies for the analysis of running. In Proceedings of the 2018 7th IEEE International Conference on Biomedical Robotics and Biomechatronics (Biorob), Enschede, The Netherlands, 26–29 August 2018; pp. 1175–1180.
- 117. Yalcin, B.; Bayram, H.A.; Bayram, M.B.; Isik, A. Investigation of Biological Feedback Influence on Posture and Knee Biomechanics During Running. In Proceedings of the 2017 21st National Biomedical Engineering Meeting (BIYOMUT), Istanbul, Turkey, 24–26 December 2017; pp. 1–4.
- 118. Bini, R.; Daly, L.; Kingsley, M. Changes in body position on the bike during seated sprint cycling: Applications to bike fitting. *Eur. J. Sport Sci.* **2020**, *20*, 35–42. [CrossRef]
- 119. Daly, L.; Bini, R.; Kingsley, M. Changes in Body Position on the Bike during Sprint Cycling: Applications to Bike Fitting. *ISBS Proc. Arch.* **2018**, *36*, 782–785.
- Di Paolo, S.; Santillozzi, F.; Zinno, R.; Barone, G.; Bragonzoni, L. On-Field Biomechanical Assessment of High and Low Dive in Competitive 16-Year-Old Goalkeepers through Wearable Sensors and Principal Component Analysis. *Sensors* 2022, 22, 7519. [CrossRef]
- 121. Wells, D.; Alderson, J.; Camomilla, V.; Donnelly, C.; Elliott, B.; Cereatti, A. Elbow joint kinematics during cricket bowling using magneto-inertial sensors: A feasibility study. *J. Sport. Sci.* **2019**, *37*, 515–524. [CrossRef]
- Nijmeijer, E.M.; Heuvelmans, P.; Bolt, R.; Gokeler, A.; Otten, E.; Benjaminse, A. Concurrent validation of the Xsens IMU system of lower-body kinematics in jump-landing and change-of-direction tasks. J. Biomech. 2023, 154, 111637. [CrossRef]
- 123. Belcic, I.; Ocic, M.; Dukaric, V.; Knjaz, D.; Zoretic, D. Effects of One-Step and Three-Step Run-Up on Kinematic Parameters and the Efficiency of Jump Shot in Handball. *Appl. Sci.* 2023, *13*, 3811. [CrossRef]
- 124. Dorsavi Ltd. Available online: https://www.dorsavi.com/ (accessed on 9 December 2023).
- 125. Charry, E.; Hu, W.; Umer, M.; Ronchi, A.; Taylor, S. Study on estimation of peak Ground Reaction Forces using tibial accelerations in running. In Proceedings of the 2013 IEEE Eighth International Conference on Intelligent Sensors, Sensor Networks and Information Processing, Melbourne, VIC, Australia, 2–5 April 2013; pp. 288–293.
- 126. Drapeaux, A.; Carlson, K. A Comparison of Inertial Motion Capture Systems: DorsaVi and Xsens. *Int. J. Kinesiol. Sports Sci.* 2020, *8*, 24. [CrossRef]
- 127. Wang, J.; Wang, Z.; Gao, F.; Zhao, H.; Qiu, S.; Li, J. Swimming Stroke Phase Segmentation Based on Wearable Motion Capture Technique. *IEEE Trans. Instrum. Meas.* 2020, *69*, 8526–8538. [CrossRef]
- 128. Incus Performance. Available online: https://www.incusperformance.com/pages/incus-nova (accessed on 9 April 2023).
- 129. Marco, B.; Giuseppe, C.; Adam, S.; Dello, I.A. The Validity and Between-Unit Variability of GNSS Units (STATSports Apex 10 and 18 Hz) for Measuring Distance and Peak Speed in Team Sports. *Front. Physiol.* **2018**, *9*, 1288.
- 130. James, R.; Mark, C.; Mikael, J.; Marco, B. Quantifying and Comparing the Match Demands of U18, U23, and 1ST Team English Professional Soccer Players. *Front. Physiol.* **2021**, *12*, 706451.
- 131. Long, C.; Jo, E.; Nam, Y. Development of a yoga posture coaching system using an interactive display based on transfer learning. *J. Supercomput.* **2022**, *78*, 5269–5284. [CrossRef]

- 132. Medtronic. Available online: https://www.zephyranywhere.com/ (accessed on 9 December 2023).
- 133. Hexoskin. Available online: https://www.hexoskin.com/collections/all/products/hexoskin-pro-kit-mens (accessed on 9 December 2023).
- 134. Haddad, M.; Hermassi, S.; Aganovic, Z.; Dalansi, F.; Kharbach, M.; Mohamed, A.O.; Bibi, K.W. Ecological Validation and Reliability of Hexoskin Wearable Body Metrics Tool in Measuring Pre-exercise and Peak Heart Rate during Shuttle Run Test in Professional Handball Players. Front. Physiol. 2020, 11, 957. [CrossRef]
- 135. Akintola, A.A.; van de Pol, V.; Bimmel, D.; Maan, A.C.; van Heemst, D. Comparative Analysis of the Equivital EQ02 Lifemonitor with Holter Ambulatory ECG Device for Continuous Measurement of ECG, Heart Rate, and Heart Rate Variability: A Validation Study for Precision and Accuracy. *Front. Physiol.* 2016, 7, 391. [CrossRef]
- 136. Stoltenberg, B.E.; Miller, E.M.; Dolbeer, J.A.; Pickens, B.B.; Goss, D.L. Validity of an instrumented sock and on-shoe sensor to provide biometric feedback to runners. *Footwear Sci.* **2019**, *11*, 147–152. [CrossRef]
- Lynn, S.K.; Watkins, C.M.; Wong, M.A.; Balfany, K.; Feeney, D.F. Validity and Reliability of Surface Electromyography Measurements from a Wearable Athlete Performance System. J. Sports Sci. Med. 2018, 14, 205–215.
- 138. Strive. Available online: https://strive.tech/ (accessed on 9 December 2023).
- Bello, M.; Anglin, D.; Gillen, Z.M.; Smith, J. The Use of Wearable Technology to Quantify Power and Muscle Load Differences During Running Against Varying Wind Resistances. *Int. J. Kinesiol. Sports Sci.* 2022, 10, 11–15. [CrossRef]
- 140. Guided Knowledge. Available online: https://dragonflygolf.guidedknowledge.com/3dgolf/smart-suit (accessed on 9 December 2023).
- 141. Komondo Technologies, Inc. Available online: https://komodotec.com/aio-specs-ekg/ (accessed on 9 December 2023).
- 142. Fleisig, G.S.; Laughlin, W.A.; Aune, K.T.; Cain, E.L.; Dugas, J.R.; Andrews, J.R. Differences among fastball, curveball, and change-up pitching biomechanics across various levels of baseball. *Sports Biomech.* **2016**, *15*, 128–138. [CrossRef]
- 143. Myontec Ltd. Available online: https://www.myontec.com/ (accessed on 9 December 2023).
- 144. Pihlainen, K.; Pesola, A.J.; Helén, J.; Häkkinen, K.; Finni, T.; Ojanen, T.; Vaara, J.P.; Santtila, M.; Raitanen, J.; Kyröläinen, H. Training-Induced Acute Neuromuscular Responses to Military Specific Test during a Six-Month Military Operation. *Int. J. Environ. Res. Public Health* 2021, 18, 215. [CrossRef]
- 145. Moov Inc. Available online: https://welcome.moov.cc/notice (accessed on 9 December 2023).
- 146. Du, K.; Lin, R.; Yin, L.; Ho, J.S.; Wang, J.; Lim, C.T. Electronic textiles for energy, sensing, and communication. *iScience* 2022, 25, 5. [CrossRef]
- Cleary, F.; Srisa-An, W.; Henshall, D.C.; Balasubramaniam, S. Emerging AI Technologies Inspiring the Next Generation of E-Textiles. *IEEE Access* 2023, 11, 56494–56508. [CrossRef]
- 148. Vieira, D.; Carvalho, H.; Providência, B. E-Textiles for Sports: A Systematic Review. JBBBE 2022, 57, 37–46. [CrossRef]
- 149. Zeng, Z.; Liu, Y.; Hu, X.; Meihua, T.; Wang, L. Validity and reliability of inertial measurement units on lower extremity kinematics during running: A systematic review and meta-analysis. *Sports Med. Open* **2022**, *8*, 86. [CrossRef]
- 150. Yeadon, M.R.; Pain, M.T.G. Fifty years of performance-related sports biomechanics research. J. Biomech. 2023, 155, 111666. [CrossRef]
- Dahl, K.D.; Dunford, K.M.; Wilson, S.A.; Turnbull, T.L.; Tashman, S. Wearable sensor validation of sports-related movements for the lower extremity and trunk. *Med. Eng. Phys.* 2020, 84, 144–150. [CrossRef]
- 152. Jenkins, L.; Weerasekera, R. Sport-related back injury prevention with a wearable device. *Biosens. Bioelectron.* X 2022, 11, 100202. [CrossRef]
- 153. Taborri, J.; Palermo, E.; Rossi, S. WARNING: A Wearable Inertial-Based Sensor Integrated with a Support Vector Machine Algorithm for the Identification of Faults during Race Walking. *Sensors* **2023**, *23*, 5245. [CrossRef]
- 154. Li, M.; Torah, R.; Nunes-Matos, H.; Wei, Y.; Beeby, S.; Tudor, J.; Yang, K. Integration and Testing of a Three-Axis Accelerometer in a Woven E-Textile Sleeve for Wearable Movement Monitoring. *Sensors* 2020, 20, 5033. [CrossRef]
- 155. Paul, G.; Torah, R.; Beeby, S.; Tudor, J. Novel active electrodes for ECG monitoring on woven textiles fabricated by screen and stencil printing. *Sens. Actuator A Phys.* 2015, 221, 60–66. [CrossRef]
- 156. Ferri, J.; Llinares, R.; Segarra, I.; Cebrián, A.; Garcia-Breijo, E.; Millet, J. A new method for manufacturing dry electrodes on textiles. Validation for wearable ECG monitoring. *Electrochem. Commun.* **2022**, *136*, 107244. [CrossRef]
- 157. Nigusse, A.B.; Malengier, B.; Mengistie, D.A.; Van Langenhove, L. A washable silver-printed textile electrode for ECG monitoring. *Eng. Proc.* **2021**, *6*, 63.
- 158. Shathi, M.A.; Chen, M.; Khoso, N.A.; Rahman, M.T.; Bhattacharjee, B. Graphene coated textile based highly flexible and washable sports bra for human health monitoring. *Mater. Des.* **2020**, *193*, 108792. [CrossRef]
- 159. Trindade, I.G.; Martins, F.; Baptista, P. High electrical conductance poly(3,4-ethylenedioxythiophene) coatings on textile for electrocardiogram monitoring. *Synth. Met.* 2015, 210 Part B, 179–185. [CrossRef]
- 160. Meghrazi, M.A.; Tian, Y.; Mahnam, A.; Bhattachan, P.; Eskandarian, L.; Kakhki, S.T.; Popovic, M.R.; Lankarany, M. Multichannel ECG recording from waist using textile sensors. *BioMed. Eng. OnLine* **2020**, *19*, 48.
- 161. An, X.; Stylios, G.K. A Hybrid Textile Electrode for Electrocardiogram (ECG) Measurement and Motion Tracking. *Materials* **2018**, *11*, 1887. [CrossRef]
- Arquilla, K.; Webb, A.K.; Anderson, A.P. Woven electrocardiogram (ECG) electrodes for health monitoring in operational environments. In Proceedings of the 42nd Annual International Conference of the IEEE Engineering in Medicine & Biology Society (EMBC), Montreal, QC, Canada, 20–24 July 2020; pp. 4498–4501.

- Brehm, P.J.; Anderson, A.P. Modeling the Design Characteristics of Woven Textile Electrodes for long—Term ECG Monitoring. Sensors 2023, 23, 598. [CrossRef]
- 164. Bystricky, T.; Moravcova, D.; Kaspar, P.; Soukup, R.; Hamacek, A. A comparison of embroidered and woven textile electrodes for continuous measurement of ECG. In Proceedings of the 39th International Spring Seminar on Electronics Technology (ISSE), Pilsen, Czech Republic, 18–22 May 2016; pp. 7–11.
- 165. Nigusse, A.B.; Malengier, B.; Mengistie, D.A.; Maru, A.; Tseghai, G.B.; Van Langenhove, L. Embroidered Textiles Electordes for Long-Term ECG Monitoring. In Proceedings of the 8th International Conference on Intelligent Textiles & Mass Customisation, Montreal, QC, Canada, 18–21 September 2022; p. 012002.
- 166. Zhao, J.; Deng, J.; Liang, W.; Zhao, L.; Dong, Y.; Wang, X.; Lin, L. Water-retentive, 3D knitted textile electrode for long-term and motion state bioelectrical signal acquisition. *Compos. Sci. Technol.* **2022**, 227, 109606. [CrossRef]
- 167. Alshabouna, F.; Lee, H.S.; Barandun, G.; Tan, E.; Cotur, Y.; Asfour, T.; Gonzalez-Macia, L.; Coatsworth, P.; Núnez-Bajo, E.; Kim, J.S.; et al. PEDOT:PSS-modified cotton conductive thread for mass manufacturing of textile-based electrical wearable sensors by computerized embroidery. *Mater. Today* 2022, 59, 56–67. [CrossRef]
- 168. Sun, F.; Yi, C.; Li, W.; Li, Y. A wearable H-shirt for exercise ECG monitoring and individual lactate threshold computing. *Comput. Ind.* **2017**, 92–93, 1–11. [CrossRef]
- 169. Ousaka, D.; Hirai, K.; Sakano, N.; Morita, M.; Haruna, M.; Hirano, K.; Yamane, T.; Teraoka, A.; Sanou, K.; Oozawa, S.; et al. Initial evaluation of a novel electrocardiography sensor-embedded fabric wear during a full marathon. *Heart Vessel.* 2022, 37, 443–450. [CrossRef]
- 170. Bosco, G.; De Marzi, E.; Michieli, P.; Omar, H.R.; Camporesi, E.M.; Padulo, J.; Paoli, A.; Mangar, D.; Schiavon, M. 12-lead Holter monitoring in diving and water sports: A preliminary investigation. *Diving Hyperb. Med.* **2014**, *44*, 202–207.
- 171. Türker, H.; Sze, H. Surface electromyography in sports and exercise. In *Electrodiagnosis in New Front Clinical Research*, 1st ed.; Turker, H., Ed.; IntechOpen: London, UK, 2013; pp. 175–194.
- 172. Adão Martins, N.R.; Annaheim, S.; Spengler, C.M.; Rossi, R.M. Fatigue Monitoring Through Wearables: A State-of-the-Art Review. *Front. Physiol.* **2021**, *12*, 790292. [CrossRef]
- 173. Ohiri, K.A.; Pyles, C.O.; Hamilton, L.H.; Baker, M.M.; McGuire, M.T.; Nguyen, E.Q.; Osborn, L.E.; Rossick, K.M.; McDowell, M.G.; Strohsnitter, L.M.; et al. E-textile based modular sEMG suit for large area level of effort analysis. *Sci. Rep.* 2022, *12*, 9650. [CrossRef]
- Liu, S.-H.; Lin, C.-B.; Chen, Y.; Chen, W.; Huang, T.-S.; Hsu, C.-Y. An EMG Patch for the Real-Time Monitoring of Muscle-Fatigue Conditions During Exercise. *Sensors* 2019, 19, 3108. [CrossRef]
- 175. Ye, X.; Shi, B.; Li, M.; Fan, Q.; Qi, X.; Liu, X.; Zhao, S.; Jiang, L.; Zhang, X.; Fu, K.; et al. All-textile sensors for boxing punch force and velocity detection. *Nano Energy* **2022**, *97*, 107114. [CrossRef]
- 176. Ma, Y.; Ouyang, J.; Raza, T.; Li, P.; Jian, A.; Li, Z.; Liu, H.; Chen, M.; Zhang, X.; Qu, L.; et al. Flexible all-textile dual tactile-tension sensors for monitoring athletic motion during taekwondo. *Nano Energy* **2021**, *85*, 105941. [CrossRef]
- 177. Molinaro, N.; Massaroni, C.; Lo Presto, D.; Saccomandi, P.; Di Tomaso, G.; Zollo, L.; Perego, P.; Andreoni, G.; Schena, E. Wearable textile based on silver plated knitted sensor for respiratory rate monitoring. In Proceedings of the 40th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), Honolulu, HI, USA, 18–21 July 2018; pp. 2865–2868.
- 178. Memarian, F.; Rahmani, S.; Yousefzadeh, M.; Latifi, M. Wearable Technologies in Sportswear. In *Materials in Sports Equipment*, 2nd ed.; Subic, A., Ed.; Woodhead Publishing: Cambridge, UK, 2019; pp. 123–160.
- 179. Mersch, J.; Winger, H.; Nocke, A.; Cherif, C.; Gerlach, G. Experimental Investigation and Modeling of the Dynamic Resistance Response of Carbon Particle-Filled Polymers. *Macromol. Mater. Eng.* **2020**, *305*, 2000361. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.